Author's response to reviews

Title: Can body mass index, waist circumference, waist-hip ratio and waist-height ratio predict the presence of multiple metabolic risk factors in Chinese subjects?

Authors:

Yong Liu (Liuy@sj-hospital.org)
Guanghui Tong (ghtong@126.com)
Weiwei Tong (tongww@sj-hospital.org)
Liping Lu (lulp@sj-hospital.org)
Xiaosong Qin (qinxs@sj-hospital.org)

Version: 3 Date: 28 December 2010

Author's response to reviews: see over
Dear editors and reviewers:

Thank you so much for accepting our manuscript entitled *Can body mass index, waist circumference, waist-hip ratio and waist-height ratio predict the presence of multiple metabolic risk factors in Chinese subjects?*. The manuscript has been revised and the next is a point-by-point description of the changes made.

Formatting changes to be made:

-----------------------------

Major revisions:

Authors' contributions: The authors’ contributions have been displayed in a separate section from the acknowledgements.

Minor revisions:

Title: Sentence cases in the title, both in the manuscript file and the submission system have been made. Capitalise only the first word and proper nouns, and do not include full stops/periods.

Text: All text is displayed in black throughout the manuscript.

Introduction: This section has been renamed as 'Background'.

Figures: The legend and title of four figures were placed as the part of the manuscript after the reference list. The figures are numbered automatically in the order in which they are uploaded.

Figures: The figures are cropped as closely as possible to minimize white space around the image.

References: The references list all authors' names for each citation.
Typography: We have checked the manuscript for typographical errors and made the corrections.

Thank you!

Address correspondence to: Yong Liu, Clinical laboratory, Shengjing Hospital of China Medical University. No. 36 San Hao Street, Heping District, Shenyang 110004, People’s Republic of China. E-mail: Liuy@sj-hospital.org.

Thank you very much.

Sincerely yours,

All authors
2010-12-28